\documentclass{article}
\input{6045preamble.tex}
\setlength{\textheight}{9in}

\begin{document}

\handout{0}{Prof. Ron Rivest}{6 Wednesday 2002}{Jonathan
  Herzog}{Handout 0: Student Survey}

Please fill in all the information and return it to the TA before the
end of the class. (If you cannot fill this in now, please return it to
the course secretary ASAP.) Please write \textbf{legibly}.

\Large

\begin{enumerate}
\item Name: 
\item Year:
\item Course/Major/University:
\item Email Address:
\item Phone:
\item Taking this class for credit?
\large
\item Have you taken 6.042?
\item If not, why do you think that you are qualified to take this
  class?
  \vspace{1in}
\item Have you had any other classes you believe to be relevant?
  \vspace{1in}
\item Why do you want to take this class?
  \vspace{1in}
\item Are you satisfying a requirement with this class? If so, what
  grade do you need? \textbf{(Note: No promises....)}
  \vspace{1in}
\end{enumerate}

\end{document}

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: t
%%% End: 
